Title: Cell Killing and Resistance in Pre-Operative Breast Cancer Chemotherapy

Authors:

Paolo P Ubezio (ubezio@marionegri.it)
David D Cameron (d.cameron@ncrn.org.uk)

Version: 3 Date: 26 March 2008

Author's response to reviews:

Milan, March 27th, 2008

Ref.: MS: 3191606721644230

Cell killing and resistance in pre-operative breast cancer chemotherapy

Paolo Ubezio and David Cameron

Dear Sir,

Thank you for forwarding us the helpful comments of the reviewers.

The requested correction has been made in the revised version (the left parenthesis in the first set of equations on p.23 have been deleted).

Thanking the referees for their constructive criticism, we hope that you find this Revised Version acceptable for publication in BMC Cancer.

In the meantime, I would like to send you my best wishes and regards.

Sincerely yours,

Dr. Paolo Ubezio
Istituto di Ricerche Farmacologiche “Mario Negri”
Via La Masa, 19
20156 Milano, Italy
tel. 0039-02-39014438
FAX 0039-02-3546277
EMAIL: UBEZIO@MARIONEGRI.IT